Author's response to reviews

Title: Factors associated with paradoxical immune response to antiretroviral therapy in HIV infected patients: a case control study.

Authors:

Janaína A S Casotti (janainacasotti@yahoo.com.br)
Luciana N Passos (lunaneves@uol.com.br)
Fabiano J P Oliveira (fabianojpo@yahoo.com.br)
Crispim Cerutti Jr (fil.cris@terra.com.br)

Version: 3 Date: 28 October 2011

Author's response to reviews: see over
Dear Sir,

Regarding the requested revisions of the manuscript, the following modifications were made:

1. The list of authors was modified as requested.
2. The tables 2 and 3 were merged.
3. All coloured text was turned black.
4. The tables titles were place at the top of them, as requested.
5. The text was revised and minor corrections were made.

I am not sure if the figures are correctly displayed. I did not succeed in accessing adequately the link for assistance in the formatting of the figures. If some other adjustment is necessary, please let me know.

Regarding payment, I could not provide it by credit card (the number was not accepted, I do not know why). So, I requested an invoice. I hope it will not present any impairment to the submission process. If it is the case, I would like to receive your instructions.

Looking forward for your future contact, I present my best regards.

Sincerely

Crispim Cerutti Junior.